\documentclass[11pt]{article}

\usepackage[margin=1in]{geometry}
\usepackage[svgnames,table]{xcolor}

\setlength{\parindent}{0pt}
\setlength{\parskip}{1ex}
\pagestyle{empty}

% question style
\newcommand{\Q}[1]{\textcolor{SlateBlue}{\textit{#1}}\medskip}

%------------------------------------------------------------------------------%
\begin{document}

\large
CS 374, HW1 \hfill Due: 01/23/2022
\begin{center}
\textbf{Type Your Name Here}
\end{center}
\hrule
\normalsize

%------------------------------------------------------------------------------%
\newpage
\subsection*{Question 1.2}
\Q{What are the key differences between the file versus database approach to data management? Please answer in two paragraphs: one that summarizes the file approach, and one that summarizes the database approach.}

Type your answer here.

%------------------------------------------------------------------------------%
\newpage
\subsection*{Question 1.4}
\Q{What are the advantages of database systems and database management? Discuss three specific examples from the end of Chapter~1 in your own words.}

Type your answer here.

%------------------------------------------------------------------------------%
\newpage
\subsection*{Question 2.2}
\Q{What is the difference between procedural and declarative DML? Please answer in two paragraphs: one that describes procedural DML, and another that describes declarative DML.}

Type your answer here.

%------------------------------------------------------------------------------%
\newpage
\subsection*{Question 2.5, Part a}
\Q{Make a list of each type of DBMS described in Section~2.2.1, and for each one, summarize in 1--2 sentences what makes it different from the others.}

Type your answer here.

%------------------------------------------------------------------------------%
\newpage
\subsection*{Question 2.5, Part c}
\Q{Make a list of each type of DBMS described in Section~2.2.3, and for each one, summarize in 1--2 sentences what makes it different from the others.}

Type your answer here.

%------------------------------------------------------------------------------%
\end{document}
